Author's response to reviews

Title: Designing an implementation strategy to improve interprofessional shared decision making in sciatica: study protocol of the DISC study

Authors:

Stefanie N. Hofstede (s.n.hofstede@lumc.nl)
Perla J. Marang-van de Mheen (p.j.marang@lumc.nl)
Willem J.J. Assendelft (w.j.j.assendelft@lumc.nl)
Carmen L.A. Vleggeert-Lankamp (c.l.a.m.vleggeert-lankamp@lumc.nl)
Anne M. Stiggelbout (a.m.stiggelbout@lumc.nl)
Patrick C.A.J. Vroomen (c.a.j.vroomen@umcg.nl)
Wilbert B. van den Hout (hout@lumc.nl)
Thea P.M. Vliet Vlieland (t.p.m.vliet_vlieland@lumc.nl)
Leti van Bodegom-Vos (l.vanbodegom-vos@lumc.nl)

Version: 4 Date: 29 May 2012

Author's response to reviews:

Dear Sir,

We agree with the changes made by the copy-editor, and have thus accepted all of these changes.

Regards,

Leti van Bodegom-Vos
(for the authors)